

Fluvial Depositional Systems of the African Humid Period: An Analog for an Early, Wet Mars in the Eastern Sahara

A. S. Zaki, J. M. Davis, K. S. Edgett, R. Giegengack, M. Roige, S. Conway, Mathieu Schuster, S. Gupta, F. Salese, K. S. Sangwan, et al.

▶ To cite this version:

A. S. Zaki, J. M. Davis, K. S. Edgett, R. Giegengack, M. Roige, et al.. Fluvial Depositional Systems of the African Humid Period: An Analog for an Early, Wet Mars in the Eastern Sahara. Journal of Geophysical Research. Planets, 2022, 127, 10.1029/2021JE007087. insu-03682337

HAL Id: insu-03682337 https://insu.hal.science/insu-03682337

Submitted on 31 May 2022 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



JGR Planets

RESEARCH ARTICLE

10.1029/2021JE007087

Key Points:

- Ancient depositional rivers in southern Egypt record ephemeral fluvial systems formed due to intense rainfall over ~10 ka
- These fluvial systems suggest tens to hundreds of years of river activity
- Similar martian systems imply that early Mars's surface was punctuated by local and transient drainage systems fed over short durations

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. S. Zaki, abdallah.bakri@unige.ch

Citation:

Zaki, A. S., Davis, J. M., Edgett, K. S., Giegengack, R., Roige, M., Conway, S., et al. (2022). Fluvial depositional systems of the African humid period: An analog for an early, wet Mars in the eastern Sahara. *Journal of Geophysical Research: Planets*, *127*, e2021JE007087. https://doi. org/10.1029/2021JE007087

Received 7 OCT 2021 Accepted 27 APR 2022

Author Contributions:

Conceptualization: A. S. Zaki, J. M. Davis, S. Gupta, S. Castelltort Data curation: A. S. Zaki, J. M. Davis, S. Conway

Formal analysis: A. S. Zaki, M. Roige Funding acquisition: S. Castelltort Investigation: A. S. Zaki, J. M. Davis, K. S. Edgett, R. Giegengack, M. Schuster, S. Gupta

Methodology: A. S. Zaki, J. M. Davis, S. Castelltort

Supervision: A. S. Zaki, J. M. Davis, K. S. Edgett, S. Gupta, S. Castelltort Writing – review & editing: A. S. Zaki,

J. M. Davis, K. S. Edgett, R. Giegengack, M. Roige, S. Conway, M. Schuster,

© 2022. The Authors. This is an open access article under the terms of the Creative Commons Attribution License, which permits use,

distribution and reproduction in any medium, provided the original work is properly cited.

Fluvial Depositional Systems of the African Humid Period: An Analog for an Early, Wet Mars in the Eastern Sahara

A. S. Zaki¹ , J. M. Davis² , K. S. Edgett³ , R. Giegengack⁴, M. Roige⁵ , S. Conway⁶ , M. Schuster⁷ , S. Gupta⁸, F. Salese^{9,10}, K. S. Sangwan⁸ , A. G. Fairén^{9,11} , C. M. Hughes¹², C. F. Pain¹³ , and S. Castelltort¹

¹Department of Earth Sciences, University of Geneva, Geneva, Switzerland, ²Department of Earth Sciences, Natural History Museum, London, UK, ³Malin Space Science Systems, Inc., San Diego, CA, USA, ⁴Department of Earth & Environmental Science, University of Pennsylvania, Philadelphia, PA, USA, ⁵Department de Geologia, Universitat Autònoma de Barcelona, Barcelona, Spain, ⁶CNRS UMR 6112 Laboratoire de Planétologie et Géodynamique, Université de Nantes, Nantes, France, ⁷Université de Strasbourg, CNRS, Institut Terre et Environnement de Strasbourg, Strasbourg, France, ⁸Department of Earth Sciences and Engineering, Imperial College London, London, UK, ⁹Centro de Astrobiología (CSIC-INTA), Torrejón de Ardoz, Madrid, Spain, ¹⁰International Research School of Planetary Sciences (IRSPS), Università d'Annunzio, Pescara, Italy, ¹¹Department of Astronomy, Cornell University, Ithaca, NY, USA, ¹²Department of Geosciences, University of Arkansas, Fayetteville, AR, USA, ¹³MED_Soil, Departamento de Cristlografía, Mineralogía y Quimica Agrícola, Universidad de Sevilla, Sevilla, Spain

Abstract A widely hypothesized but complex transition from widespread fluvial activity to predominantly aeolian processes is inferred on Mars based on remote sensing data observations of ancient landforms. However, the lack of analysis of in situ martian fluvial deposits hinders our understanding of the flow regime nature and sustainability of the martian fluvial activity and the hunt for ancient life. Studying analogs from arid zones on Earth is fundamental to quantitatively understanding geomorphic processes and climate drivers that might have dominated during early Mars. Here we investigate the formation and preservation of fluvial depositional systems in the eastern Sahara, where the largest arid region on Earth hosts important repositories of past climatic changes. The fluvial systems are composed of well-preserved single-thread sinuous to branching ridges and fan-shaped deposits interpreted as deltas. The systems' configuration and sedimentary content suggest that ephemeral rivers carved these landforms by sequential intermittent episodes of erosion and deposition active for 10-100s years over ~10,000 years during the late Quaternary. Subsequently, these landforms were sculpted by a marginal role of rainfall and aeolian processes with minimum erosion rates of 1.1 ± 0.2 mm/yr, supplying ~96 ± 24×10^{10} m³ of disaggregated sediment to adjacent aeolian dunes. Our results imply that similar martian fluvial systems preserving single-thread, short distance source-to-sink courses may have formed due to transient drainage networks active over short durations. Altogether, this study adds to the growing recognition of the complexity of interpreting climate history from orbital images of landforms.

Plain Language Summary Mars is currently a dry and cold desert, but rivers preserved in inverted topography suggest that water once flowed during its early history. However, how sustained and how frequently these rivers flowed remains uncertain. Here we study ancient fluvial systems (rivers and deltas) from the eastern Sahara that formed during the late Quaternary, in much wetter conditions than those prevailing today in this desert and which bear striking analogies to martian systems. We find that rivers and deltas, now preserved as ridges, record short distance source-to-sink high-energy systems formed due to heavy rainfall events. Our observations and measurements of the meandering systems within the deltaic features suggest that such wet conditions might have spanned tens to a few hundred years over a total duration of ~10,000 years. Since the wet conditions ceased, arid conditions prevailed, and the aeolian processes resumed, sculpting ridges out of ancient channels. Our results imply that martian fluvial systems may have been associated with similar local and heavy runoff conditions that lasted 10–100s years over thousands of years, possibly sufficient to support habitability. A shift toward arid environments led to the sculpting of fluvial ridges and the widespread formation of dunes across the modern martian landscape.



F. Salese, K. S. Sangwan, A. G. Fairén, C. M. Hughes, C. F. Pain, S. Castelltort

1. Introduction

Mars is currently hyperarid, but considerable geological evidence suggests that the martian surface was at least episodically warm during the late Noachian and early Hesperian ca. 3–4 Ga (e.g., Malin & Edgett, 2003; Fassett & Head, 2008; Davis et al., 2016; Salese et al., 2019; Wordsworth, 2016; Kite, 2018; Dickson et al., 2020). This warming epoch might have triggered both ice melting and hydrological cycle activation, forming valley networks, paleolakes, and fan-shaped deposits (e.g., Di Achille & Hynek, 2010; Goudge et al., 2012, 2018; Gulick, 2001; Kite, 2018; Mangold et al., 2004; Wordsworth, 2016). Following these periods, Mars transitioned from wet-to-dry conditions, likely driven by reductions in atmospheric pressure (Lammer et al., 2013), although the precise nature of this transition is unclear (Kite et al., 2019). Since the Hesperian, the martian landscape has been exposed to prolonged aeolian erosion, revealing much of the planet's ancient geologic record. Former stream deposits (at all scales from rill to creek to river) are now exposed as ridges at numerous locations across the martian surface in a variety of settings (e.g., Burr et al., 2010; Cardenas et al., 2017; Davis et al., 2019; Di Pietro et al., 2018; Dickson et al., 2020; Williams, 2007; Zaki, Pain, et al., 2021). Some of these ridges have been used to reconstruct the paleohydrology and paleoclimate of early Mars (Burr et al., 2010; Davis et al., 2016; Kite et al., 2019; Hayden, Lamb, & McElroy 2021).

Such ridges on Earth result from a cyclic process in which ancient rivers were filled with resistant sediment, including volcanic products, large clasts, and/or cemented clasts, and were subsequently topographically inverted by differential erosion to stand as ridges in the modern landscape (e.g., Maizels, 1983, 1987, 1990; Williams et al., 2007, 2009, 2021; Foix et al., 2012; Madof et al., 2019; Pain & Ollier, 1995; Zaki, Pain, et al., 2021). In some cases, such ridges appear in the landscape after deep burial, followed by exhumation and net landscape lowering by differential erosion (e.g., Balme et al., 2020; Cardenas et al., 2020; Davis et al., 2019; Hayden et al., 2019; Ielpi & Ghinassi, 2014; Zaki, Pain, et al., 2021). Ridges are observed in different forms, including flat-topped, elongated sinuous ridges, fan-shaped deposits (deltas and fluvial fans), meanders, and point bars (Figure 1; Davis et al., 2016; Cardenas et al., 2020; Hayden et al., 2019; Hayden, Lamb, & Carney, 2021; Hayden, Lamb, & McElroy, 2021; Dickson et al., 2020; Zaki, Pain, et al., 2021). The investigation of these landforms on Mars is now only possible through remotely sensed data from orbiters (although this might soon change, at least in Jezero crater, when the Perseverance rover arrives at the deltaic landform in Figure 1d; Mangold et al., 2021), hindering our ability to measure their exact geometries, understand their development, how much time they record, and the mechanisms of their preservation (e.g., sediment cementation) during the shift from wet to arid environments. Although fluvial ridges on Earth have been widely investigated (e.g., Pain & Ollier, 1995; Williams et al., 2007, 2009, 2021, Clarke et al., 2020; Hayden et al., 2019; Hayden, Lamb, & Carney, 2021; Hayden, Lamb, & McElroy, 2021; Zaki et al., 2018; Zaki, Pain, et al., 2021, further studies of similar landforms in dry climates on Earth help provide uniformitarian analogs of the physical principles of landform evolution processes, including streamflow mechanisms, duration of fluvial activity, patterns of fluvial deposition, processes of relief inversion, and the nature of the transition from wet to arid climates. Constraining such information can refine our understanding of martian landforms with similar planforms and sedimentary structures by providing new insights into the paleoenvironmental conditions associated with their development. Such information may also improve our understanding of habitability and potential biosignature on early Mars.

The Sahara Desert, the largest dry region on Earth, contains superb repositories of past climatic and environmental changes from wet to arid conditions recorded by multiple geologic proxies. The record of the fluvial activity has been preserved as fossilized fluvial landforms due to the hyper-arid conditions. A snapshot of this record is well-preserved in southern Egypt, where ancient riverbeds and fans survive as inverted topography in the modern landscape. These landforms occur over an ~38,000 km² region on the western and eastern sides of Lake Nasser (Figure 2; Zaki & Giegengack, 2016; Giegengack & Zaki, 2017). These landforms are well-preserved in inverted topography, taking different forms, including fine-branched ridges, single-thread ridges, and sedimentary fans (bearing inverted channel forms). Clastic sediments preserved in these landforms are well-cemented (e.g., Giegengack & Zaki, 2017). After cementation, the sediments were left standing as ridges by a regional lowering of the desert surface due to aeolian and fluvial erosion during a period of climatic oscillation (Giegengack & Zaki, 2017). The discovery of Acheulian artifacts and pottery shards within the fluvial deposits stacked in the ridges established their relative age from middle Pleistocene to Holocene (Giegengack & Zaki, 2017). Furthermore, Zaki, King, et al. (2021) recently dated these paleo-rivers using OSL and ¹⁴C; their results indicate that they were formed by several depositional events ranging from 53 ka to 1 ka. The ages indicate intense fluvial activity





Figure 1. Topographically inverted landscape features: (a) A short-distance source-to-sink paleo-river, now expressed as a set of ridges and a fan in the modern landscape, in the western Medusae Fossae Formation region of Mars. The entire source to sink fluvial system stands as ridges in the modern landscape (CTX image P21_009109_1797_XN_00S213W; 0°15'52.24"S; 146°25'26.07"E). (b) Deltaic deposit in Eberswalde crater that preserves many topographically inverted paleochannels (CTX image P01_001534_1559_XI_24S033W; 23°49'9.76"S; 33°39'45.97"W). (c) Fluvial ridges along on the plateau west of Juventae Chasma (CTX image G06_020602_1757_XN_04S063W 4°14'35.67"S; 63°14'34.09"W). (d) Perspective view of the Jezero crater "western delta" showing ridges interpreted by (Goudge et al., 2018) to be inverted channels (Credit: NASA/JPL/Seán Doran; 18°33'19.90"N; 77°23'36.79"E).

during the early (~13,000 BP) to middle (~5,200 BP) Holocene driven by heavy rainfall events (Zaki, King, et al., 2021). The intensity of rainfall events was briefly calculated using grain size and paleo-channel geometry to obtain the discharge, then dividing the discharge by drainage area leads to minimum precipitation rates (Zaki, King, et al., 2021). This intense episode of humid conditions has been linked to the African Humid Period - AHP (ca. 14.8 – ca. 5.5 ka BP; deMenocal et al., 2000; Hoelzmann et al., 2000; Nicoll, 2004; Schuster et al., 2005; Palchan & Torfstien, 2019). The AHP is an episode of increased insolation due to orbital changes, leading to a rise in temperatures by 7.5°C and precipitation by 3–4 times more before the AHP, turning the Sahara into a savannah-like environment (Figure 3; deMenocal et al., 2000; Martrat et al., 2004; Kuper & Kropelin, 2006, Hoelzmann et al., 2000). After the termination of the AHP, the wet conditions abruptly ceased; this region today receives <2 mm per year of rainfall; thus, aeolian processes prevail (Hamdan et al., 2015; Nicoll, 2004).

In this contribution, we address the fundamental issues outlined above by investigating a diverse array of fluvial landform remnants, which represent a natural laboratory to quantitively understand the duration and the nature of the flow regime associated with similar martian features. We compiled remotely sensed data, field observations and measurements, petrographic observations, empirical relations, and published chronological data to reconstruct these landforms' formation, preservation, and evolution through time. These reconstructions provide new insights into the martian fluvial history, which may have undergone a similar drying to the Quaternary Sahara.

2. Regional Setting

The southeastern part of the Western Desert of Egypt (the eastern Sahara) covers an area of ~50,000 km². The landscape includes vast plains, cuestas, small basins, sporadically distributed hills, and aeolian dunes. The region can be broken into three physiographic units (Hamdan et al., 2016), the Sinn el-Kiddab Plateau, the Tushka Plain, and Lake Nasser. The Sinn el-Kiddab Plateau is underlain by horizontal limestone and shale units that were deposited during late Cretaceous to early Eocene time (Figure S1 in Supporting Information S1). This plateau now stands at an elevation of 300–400 m and overlooks the western side of the Nile. (Equation 2) The Tushka Plain lies south of the Sinn el-Kaddab Plateau, borders Lake Nasser to the east, and extends to Sudan in the south. This ~13,150 km² plain is underlain by the Nubia Formation, primarily quartz-rich sandstones of mid-to-late Cretaceous age; this terrain is partially overlain by a veneer of ephemeral playa lakes and associated sedimentary features that have characterized this landscape during wetter conditions during the late Quaternary time (e.g., Maxwell et al., 2010;

Nicoll, 1998, 2004; Wendorf & Schild, 1980). Lake Nasser covers \sim 5,200 km² and has the capacity to hold \sim 165 km³ of water.

This paper reconstructs the configuration of the former river systems that drained the Tushka Plain. Sediment deposited in these rivers now stands as ridges in the landscape (e.g., Giegengack & Zaki, 2017). Most of these paleorivers were developed on consolidated sedimentary quartzite, the "Nubia Formation", deposited during the late Cretaceous time (Said & Issawi, 1964; Zaki & Giegengack, 2016). The Nubia Formation consists of detrital quartzite, derived mainly from the African Shield (Giegengack & Zaki, 2017). The grain size of clasts in this formation is highly variable, ranging from cobbles in the coarse basal conglomerates to silt and clay in the finer mudstone layers (Giegengack & Zaki, 2017). The Nubia Formation unconformably overlies an igneous and metamorphic complex (Giegengack & Zaki, 2017). The Nubia formation in the region was buried, lithified, then exhumed. Tewksbury et al. (2017) estimated the burial depth of the Nubia Formation at this region to fall in the





Figure 2. (a) Locations of fluvial ridges and paleodeltaic landforms in the eastern Sahara mosaic of Landsat 8 images acquired in 2018 overlaid on a shaded relief map produced from Shuttle Radar Topography Mission (SRTM). The paleorivers discussed in this paper are; (1) Gabal Hamam I, (2) Gabal Hamam II, (3) Gabal Masmas, (4) Gabal El-Sadd, (5) Arqin I, and (6) Arqin II. (b) Example view of a dendritic fluvial ridge system, located at 22°37'36.56"N; 31°47'33.18"E (Portion of Esri World Imagery product created in 2009). (c) Example of fan-shaped landform and inverted paleochannels (paleodelta) at the western shore of dry lake Kiseiba-Tushka at 22°48'31.49"N; 30°46'56.48"E (Portion Esri World Imagery product created in 2009).

range of \sim 400– \sim 800 m. Then, it has been exhumed by tectonic uplift, coupled with a regional differential erosion (Tewksbury et al., 2017).

We also used remotely sensed data for a preliminary investigation of fan-shaped deposits that accumulated at the margins of a \sim 180 km² paleolake basin and are now preserved, in part, as inverted topography. This paleolake basin, Kiseiba-Tushka, occurs \sim 40 km west of the fluvial-ridge sites. Presently (2022), Kiseiba-Tushka is filled with water diverted from Lake Nasser via the Wadi Tushka as a result of recent Nile flooding. The Kiseiba-Tushka paleolake basin is one of a suite of shallow closed depressions, primarily wind-deflated, in southern Egypt and northern Sudan that held shallow playa lakes during a wet episode that lasted from early to mid-Holocene time (Nicoll, 1998). The paleolake basin hosts well-preserved catchment areas, fans, and overlying aeolian sand dunes, and thus it records a hydroclimatic transition from wet to arid conditions.

The southeastern part of the Western Desert region is hyperarid, presently receiving <2 mm/yr of rainfall. However, regional climatic records document extreme events of heavy rainfall that occur sporadically every 10–30 years over the Egyptian Sahara, with rainfall as much as tens of millimeters in one day (Embabi, 2018). The wind frequently blows from the north-northwest across northern Sudan and southern Egypt at speeds ranging from 4 to 20 km/hr (Oliver, 1965).

3. Materials and Methods

Unhindered by vegetation because of the hyperarid setting, we used a set of remotely sensed data (particularly satellite images) to identify and measure the geomorphic features of the fluvial ridges and the deltaic landform in the study region. These data include very high spatial resolution (\sim 50 cm/pixel) visible wavelength images obtained from the Pléiades satellite constellation. We also examined a combination of freely available satellite and aerial images presented at 2.5–15 m/pixel from Esri World Imagery and Bing Imagery, multispectral Landsat 8 scenes satellite at a scale of \sim 30 m/pixel. Declassified CORONA satellite images were used to characterize geomorphic features now submerged beneath Lake Nasser. We also employed digital elevation data from the





Figure 3. Environmental proxies from the eastern Sahara show an increase in insolation that promoted higher temperature, leading to wetter conditions between 14.8 and 5.5 ka BP (the AHP). (a) Mean annual precipitation during the AHP as reported from Selima Oasis (Haynes et al., 1989), West Nubia Paleolake Basin (Hoelzmann et al., 2000), Bir Atrun and Oyo (Ritchie & Haynes, 1987), and Lake Yoa, Tibesti and Coastal Libya (Blanchet et al., 2021), and Lake Ohrid (Wagner et al., 2019). (b) Sea surface temperature reconstructing from the Alboran Sea (Martrat et al., 2004). (c) Summer insolation at 30°N (Berger & Loutre, 1991). Increasing the insolation during the AHP was driven by orbital changes.

Japanese Advanced Land Observation Satellite (ALOS) with a spatial resolution of ~ 28 m and a vertical resolution of 5 m to extract the drainage areas, river profiles, and slopes.

Zaki, King, et al. (2021) focused on obtaining absolute ages using OSL and ¹⁴C, and reconstructing precipitation rates involved in forming the upstream paleochannels of these fluvial systems based on grain size. However, investigating the morphology and the drainage areas from the source to sink systems, the internal architecture of the whole ridge, reconstructing discharge using channel width, and calculating the inversion rates remain unconstrained. Such parameters are essential for martian ridges, as we do not see the paleochannel geometries and grain size distribution from orbiter data; we instead see ridges.

Our fieldwork included measurement of fluvial ridge heights, widths, and thicknesses of stratigraphic sections deposited in the paleorivers at 16 locations along the six ridge systems (Figure 4; Table S1 in Supporting Information S1). We also observed and documented the sedimentary structures and facies within the ridge-sediment sequences, including the storeys stacked in the ridges, imbrication, and bedding. In order to measure the grain size of these paleo-rivers, we followed a method established by Wolman (1954) to determine the distribution of materials in coarse-bedded streams. At each location, grain size data were obtained by measuring the b-axes of 101–124 clasts of >2 mm size that were randomly picked out of a 1 m² area following Wolman (1954). That is followed by determining the median of the grain size (D_{50}) at each location based on multiple measurements.





Figure 4. Satellite images of the six fluvial ridge sites showing the 16 sites we investigated along the paleo-rivers (Credit: Bing Imagery). The white stars represent the examined sites. (a) Gabal Hamam I and II sites, (b) Gabal Masmas, and (c) Gabal El-Sadd, and (d) Arqin I and II sites. The images also show that the ridges bear the dendritic pattern with their flow direction. The catchment areas are well-expressed in portions B and C, where there are the Gabal Masmas and the Gabal El-Sadd paleorivers.

We used satellite images to measure the lengths of the longest mainstream (L) of each of the fluvial ridge systems from the source to the sink to reconstruct the paleo-drainage area (A) of each paleoriver via applying an empirical relationship developed by (Hack, 1957) (Equation 1) to estimate the area of the river basin. However, Som et al. (2009) suggested that the Hack's exponent (0.5–0.6) might lead to uncertainty in drainage area calculations in the case of ephemeral rivers. We, therefore, compared our drainage area estimates to modern rivers surviving under different fluvial settings, including sustained and flashy depositional systems (e.g, Patton & Baker, 1976; Church & Rood, 1983; Trampush et al., 2014). The flashy depositional environments are typically caused by intense rainfall over short-duration events on a fast-responding landscape (Patton & Baker, 1976). This contradicts the sustained depositional settings that required large drainage areas under enhanced rainfall.

$$L = 1.4A^{0.6} \tag{1}$$

Zaki, King, et al. (2021) used paleochannel geometries, particularly channel depth and width and grain size measurements, to calculate the paleoslope, paleovelocity, and paleodischarge of the ridges in southern Egypt. However, this approach is not feasible on Mars since most of the data about fluvial ridges is obtained from orbiter data. To reconstruct the discharge, we used a width-based equation (Equation 2) developed by (Eaton, 2013), applied and showed high accurate estimates on Earth by Jacobsen and Burr (2018), and scaled to martian gravity by Kite et al. (2015) to estimate the discharge and the difference between using the two estimates. This might be helpful for a better understanding of paleohydaulics of similar features on Mars. Recent studies show that ridge width does not reflect the original channel width, and it should be larger in the case of the presence of ridges preserving channel belts (Hayden et al., 2019; Hayden, Lamb, & Carney, 2021). The rivers discussed in this paper, however, record channel fill with an absence of vertical aggradation and lateral migration. The ridge width, therefore, is nearly equal to the channel width, particularly in this type of stratigraphic architecture (e.g., Hayden et al., 2019). However, the ridges investigated in this study record channel fill that closely preserve the original channel shape (Hayden et al., 2019).

$$Q = 0.10W^{1.866} \tag{2}$$

Our petrographic interpretations were based on a study of 31 thin-sections. (2.3*4 cm) under a petrographic microscope for a better understanding of the paleoenvironmental conditions associated with the source rocks. The environmental information that can be obtained from thin-sections are the nature of deposition, cementation processes, and burial conditions. The environmental conditions can be inferred from the type of mineral, the shape and the size of the grain, and the features that were overprinted on the rocks due to differential processes throughout time (Mackenzie et al., 2017).

We calculated migration rates for meanders in the Kiseiba-Tushka paleolake basin deltaic landform using an empirical relation developed by Lapôtre and Ielpi (2020) through a global compilation of channel migration rates of the channel width (*w*) of both vegetated and unvegetated rivers. We applied this approach to the equation of vegetated rivers (Equation 3), assuming—based on the pollen record in the eastern Sahara during the AHP (Haynes et al., 1989; Ritchie & Haynes, 1987)—that the studied rivers were formed under vegetated environments. To get the duration we multiplied the migration distance by the migration rate to constrain how many years of bankful flows were required to form the meander belt.

$$Mr_{Ev} = (0.023 \pm 0.001)w^{0.85} \tag{3}$$





Figure 5. Field photographs acquired in August 2018 for the six paleorivers: (a) Gabal Hamam I (22°56'15.76"N; 32° 7'20.04"E), (b) Gabal Hamam II (22°55'17.51"N; 32° 4'25.82"E), (c) Gabal Masmas (22°51'15.28"N; 31°55'29.56"E), (d) Gabal El Sadd (22°37'30.86"N; 31°47'7.19"E), (e) Arqin I (22° 2'26.24"N; 31°12'29.75"E), and (f) Gabal Arqin II (22° 2'32.89"N; 31°16'41.35"E).

Differential erosion has lowered the desert surface to produce these ridges. We can estimate the minimum erosion rate (η) at each ridge by dividing the ridge height (*H*) by the age (*T*) of the youngest fluvial sediment in that ridge (Wang et al., 2015) (Equation 4). The ages of these ridges were measured using ¹⁴C and optically stimulated luminescence (OSL) (Zaki, King, et al., 2021), but we also used the ages derived from other proxies, including lake sediments and levels, pollen, and dust fluxes (Hoelzmann et al., 2000; Palchan & Torfstien, 2019; Schuster et al., 2005) to define when the aridification began.

$$\eta = \frac{H}{T} \tag{4}$$

4. Results

4.1. Fluvial Ridges on the Tushka Plain

4.1.1. Morphology

The six ridge systems investigated here (Figures 4 and 5) exhibit dendritic drainage patterns in plan view. They are discontinuous, flat or convextopped ridges from ~ 1.5 to ~ 13 m high, and their widths do not exceed 65 m (Figure 5; Table S1 in Supporting Information S1). The ridges were sculpted by either water or wind, but the fluvial washes and the modern drainage lines suggest that they were segmented due to rainfall and surface runoff erosion (Figure 6). The sinuosity of the ridges ranges from 1.1 to 1.4. Most of the ridges record single-thread channels without any evidence of existing point bars or braided stretches. The geometry of the surviving remnants of these stream systems suggests they record local drainage networks that collected water and sediment from small watersheds drained into the Nile, except for the Gabal El-Sadd paleoriver, which debouched into a low-relief plain on which is preserved a small fan-shaped deposit (Figure 7 and Figure S2b in Supporting Information S1). The small-scale geometry of this fan suggests a lack of significant sediment transport and the intermittency of the fluvial events (Figure 7). The distribution of the fluvial ridges is restricted to the areas of existing outcrops of sandstone of the Nubia Formation, suggesting that that unit is the only surficial rock unit that enables the required sequence of geologic processes to occur. The distance from the source to the sink is relatively short, ranging from ~6 to ~27 km in the six examples studied. Widely separated, poorly developed, and small drainage areas characterize the fluvial ridges, estimated empirically to range from 12.3 to 139 km². Based on ridge widths and using (Equation 2), we estimate the discharge at 16 locations along with the six ridge sites at 33.4 ± 9.7 m³/s and does not exceed $245 \pm 71 \text{ m}^3/\text{s}$ (Table S1 in Supporting Information S1). When we compared our paleorivers discharge to similar values from modern rivers surviving under different environments, we found that such discharge values require similar drainage areas under intense flash floods and larger drainage areas formed by sustained rainfall (Figure 8).

4.1.2. Internal Architecture

Through investigation at 16 locations along the six fluvial ridges (Table S1 in Supporting Information S1; Figure 4), we find that the ridges are composed

of two stratigraphic units. The basal unit is the Cretaceous Nubia Formation; which is overlain by Pleistocene and Holocene sandy gravels (Figures 9 and 10).

The Nubia Formation underlying the ridges consists of 0.75–12 m vertical thickness of simple cross-bedded and planar-bedded sedimentary quartzite derived from igneous and metamorphic rocks of the African Shield.





Figure 6. Oblique views show how the ridges were segmented by recent drainage lines responding to erosion by rainfall and surface runoff; (a) and (b) Gabal Hamam I paleoriver. (c) Gabal Hamam II. The images were derived from Esri World Imagery.

The clasts range in size from very fine gravel (~3 mm) to silt. The roundness of the particles varies from well-rounded to sub-rounded. Most of the unit is cemented by iron oxide, with a marginal contribution of the calcite, responsible for variable resistance to erosion. The clear sedimentary structures displayed in many outcrops, coupled with particle roundness and their petrography, support the inference that the Nubia Formation in this area is well-developed alluvial sediment consistent with the interpretation of McKee (1963), who pointed out that the Nubia Formation sandstone accumulated in Upper Cretaceous time as a floodplain and lacustrine deposit. Petrographic observations of this unit (Figure 10) do not show evidence of significant compaction, including fractures and breakages, supporting the inference that the rock unit was not deeply buried, but it might have been affected by shallow burial.

The sedimentary structure of the sandy gravels and the substrate suggests that these ridges represent stream channels incised into Nubia Formation sandstone. The gravel bodies are poorly sorted, ranging from 0.65 ± 0.2 to 1.2 ± 0.4 m in height. The gravel unit in each of the six fluvial ridges is not bedded except at Arqin II (Figure 9). The structureless attribute (lack of bedding) in most of the ridges might be relevant to the ephemeral streams, coupled with the rapid nature of the transport rates that occurred during the late Quaternary climate oscillation where there were rapid alternating episodes of the wet and dry conditions. The median grain size (D_{50}) ranges from 21 ± 2 mm to 64 ± 2 mm (coarse to very coarse gravel). The roundness of the gravel greatly varies from well-rounded to very angular, reflecting a wide range of different environments of transport, ranging from slower transportation rates such as in Argin II to more extreme rates as in the rest of the ridges. The gravels in the fluvial ridges were accumulated vertically with no evidence of lateral migration, reflecting channel fill as a stacking pattern. The sandy gravels are strongly cemented by iron oxide and calcite (Figure 11). The ages obtained by Zaki, King, et al. (2021) enabled us to calculate the annual accumulation of sediments and their response to the increased precipitation rates in Figure 3 at four rivers, including Gabal Hamam II, Gabal Masmas, Gabal El Sadd, and Argin I. We find that they range from 8×10^{-3} to



Figure 7. A trace of fluvial ridges in Gabal El-Sadd paleoriver shows that these ridges flowed a short distance from source to sink. Because the sink is now submerged beneath the water of Lake Nasser, we coupled CORONA image (ID: ds1107-1074da120) with Esri world imagery.





Figure 8. Values for drainage areas for both modern rivers survive under sustained and flashier fluvial environments and the Saharan paleorivers formed during the AHP. The figure shows that smaller drainage areas with relatively similar discharge tend to be formed under intense flash floods and the larger ones require sustained fluvial flows. The sustained rivers were compiled from (Church & Rood, 1983; Trampush et al., 2014) and flashier rivers (Patton & Baker, 1976).

 1.25×10^{-1} mm/yr. These rates peaked during the AHP due to the increased precipitation (Figure 12).

4.1.3. The Preservation of Fluvial Ridges

After the cementation of the fluvial sediments either by iron oxide or calcite, differential erosion eroded the adjacent slopes, leaving the channel floors standing as ridges. Here we estimate the erosion rates and denudation volumes that eroded the substrate materials. These estimates are minimum rates and volumes, as the reconstruction of the precise elevation of the former eroded surface is not possible with the fluvial ridges. In order to constrain a reasonable age for the regional aridification that represents the onset of processes that lowered the surface, we evaluated ages determined from different proxies, including fluvial ridges, pollen records, dust fluxes, lake levels, lake sediments, and human occupation (e.g., Kuper & Kropelin, 2006; Nicoll, 2004; Zaki, King, et al., 2021). We find that the recent aridification started at ~5 ka BP. This age is consistent with the termination of the AHP at 5.2 ka BP (e.g., deMenocal et al., 2000).

Dividing the fluvial ridge thickness at each of the six sites by the duration of aridification (~5.2 ka) yields minimum erosion rates that range from 2 ± 0.3 mm/yr at the Gabal El-Sadd site to 0.3 ± 0.1 mm/y at the Arqin site, with an average of 1.1 ± 0.2 mm/yr (Figure 13). The calculated volumes of eroded sediment are in the range of ~96 ± 24 × 10¹⁰ m³, based on multiplying the average height of the fluvial ridges (5.6 ± 1.4 m) by the area in which the ridges are distributed in the west bank of the lake (17,157 km²). Deflation by wind and the marginal role of recent rainfall have reworked these volumes of sediment to construct various types of dunes.

4.2. Fan Systems in the Kiseiba-Tushka Paleolake Basin

Our preliminary investigation from the remotely sensed data shows that twenty fan-shaped forms that terminate in the \sim 35 m-deep Kiseiba-Tushka depression, which we have been mapped (Figure 14). They are concentrated on the north and west sides of the depression, where the catchment areas and the accommodation space were sufficient to form such fans. Some parts of these fans are relatively eroded, standing in relief in the modern landscape. Planform views show structures indicative of meanders, point bars, and splay deposits (Figures 15 and 16). Such features commonly form within floodplains. Furthermore, we identified channels bordered by levees, which could reflect development within a waterbody during base-level changes (Figures 14b–14d). The low-relief depressions in the region, which encapsulates all of the features, was also identified by Nicoll (1998) as a playa lake that was likely active during the early to mid-Holocene time. Thus, we consider these fans-shaped deposits to be deltas rather than alluvial fans.

These deltas have small surface areas ranging from 0.16 to 1.71 km² (Table S2 in Supporting Information S1). Using ALOS PALSAR DEM, we obtained their present-day slopes that do not exceed 0.014, and each presents a radial fluvial distributary pattern. When considering the varying elevations of the fan deposits, we see three distinct elevations at which the slope changes in the measured long profiles of the deposits. This indicates that these deltas fanned out into the paleolake at three different lake levels which are today at 136 \pm 0.75 m asl, 141.5 \pm 0.5 m asl, and 153.5 \pm 1.5 m asl (Figures 14b and 14c). Based on measurements of the deltas' thicknesses from DEMs, we find that the best estimate ranges 10–17 m. These deltas accumulated at the base level of the lakes from drainage areas ranging from 0.18 to 4.78 km², consistent with Hack's law calculation. The trunk valleys of each catchment zone have sharp walls with theater-head valleys in most examples, suggesting post-incision modification by groundwater seepage, similar to those are concentrated along the Libyan escarpments (Abotalib et al., 2016). The valley trunks are tens of meters wide to a maximum of ~200 m.

The sedimentary bodies of each delta preserve incised small streams, fluvial ridges (topographically inverted stream channel forms), and point bars (Figures 15 and 16). In 14 of the 20 deltas, point bars are dominant. They appear in planform as sets of meander scroll ridges with individual heights ranging from >1 m to <4 m.



Journal of Geophysical Research: Planets

10.1029/2021JE007087



Figure 9. Vertical lithological sections of six of the fluvial ridges. Gabal Hamam I (22° 56′ 00″N; 32° 08′ 24″E), Gabal Hamam II (22° 54′ 39″N; 32° 04′ 13″E), Gabal Masmas (22 50 56.99N; 31 54 58.01E), Gabal El Sadd (22° 37′ 30″N; 31° 47′ 08″E), Arqin I (22° 01′ 52″N; 31° 12′ 19″E), and Arqin II (22° 02′ 33″N; 31° 16′ 39″E). The red dotted lines of the river sketches represent the missing sections.

The lateral migration distances have been measured on 48 point bars and indicate 4.6–56.3 m based on the extensive downstream stacking of adjacent meander scrolls (Table S3 in Supporting Information S1). Using an empirical relationship developed by (Lapôtre & Ielpi, 2020), the migration rate ranges from 0.06 ± 0.002 m/yr to 0.63 ± 0.03 m/yr under vegetated environments, increasing with the channel width. Dividing the migration distance by the migration rate suggests a migration time span of ~25.2 ± 0.9 to ~187 ± 6.2 years. Crevasse splay features are observed near a meandering system in one delta (Figures 15 and 16), suggesting that the water and sediments broke the outer banks during high discharge events.

Observations from satellite images for the modern analogs from Razazza Lake in Iraq and modern Tushka lake are consistent with the estimated durations, confirming that similar deltaic features require a few tens of years of continuous alluvial activity in shallow depositional environments (Figure 17). Movies S1 and S2 in Supporting Information S1, available as supporting information, also show the formation of similar deltaic features within modern lakes.

We observed many ridges that we interpret as fluvial deposits within 15 of 20 deltas (Figure 16). These ridges are well-preserved at the distal end of the fans with a lateral extent of tens of meters to a few hundred of meters. They stand up to \sim 5 m high in the modern landscape, with a maximum width of 150 m and a surface slope toward the lake. These ridges record channel belts based on their planform shape, reflecting lateral migration in only three deltas, with evidence of ridge superposition that might express vertical aggradation.





Figure 10. Field photographs showing examples of sedimentary structures of the fluvial ridges: (a) and (c) A side view of the Gabal Hamam I site $(22^{\circ} 56' 00''N; 32^{\circ} 08' 24''E)$. (b) Arqin I $(22^{\circ} 01' 52''N; 31^{\circ} 12' 19''E)$. (d) Gabal Hamam II $(22^{\circ} 54' 39''N; 32^{\circ} 04' 13''E)$. (e) Arqin II $(22^{\circ} 02' 33''N; 31^{\circ} 16' 39''E)$. (f) A front view of the Arqin II; ruler of 50 cm length for scale $(22^{\circ} 02' 33''N; 31^{\circ} 16' 39''E)$. The photographs show that most of the ridges consist of two stratigraphic units; the Nubian Formation (Cretaceous) overlaid by coarse clasts (Quaternary). The yellow dashed lines in the photos are the contact between the Nubian Formation and the coarse clasts materials that were deposited during the late Quaternary. The boundary was set based on the information reported about the Nubia Formation composition.





Figure 11. Photo micrographs showing variability in the microfacies (purple arrows indicate iron oxide and blue arrows indicate calcite; blue epoxy in pore spaces). (a) and (b) Moderately and poorly sorted sandstone with low porosity and coarse sands and gravels. (c) Moderately sorted, medium sand-sized sandstone, with sub-angular shape. (d) Well sorted sandstone, with fine-grain size, ranging from angular to sub-angular in the shape. The concretion of iron-bearing cement likely indicates a fluvial environment. (a) and (b) collected from the late Quaternary facies from Gabal Hamam I and Gababl El-Sadd sites, respectively. (c) and (d) represents different facies of the Nubia Formation from the Gabal Hamam I.

5. Discussion

The eastern Sahara' fluvial systems described here provide a unique window to understand the fluvial history that controlled the eastern Sahara during more humid conditions. They preserve short-distance locally sourced systems and consist of fluvial ridges and the deltaic features, recording a series of climatic-driven landscape changes before, during, and after the AHP. These systems were formed by channel-belt and channel-fill construction by river-channel migration and aggradation. Channel incision and sediment deposition were followed, during more arid times, by topographic inversion via differential erosion by wind and erosion by rainfall-generated runoff. Such landforms in arid regions are instructive for our understanding of the early martian climate, as they are a snapshot of fluvial history followed by aeolian environments. Thus, the fluvial landscape in southern Egypt provides a natural laboratory to consider the paleoclimate and paleoenvironment context for the evolution of similar martian landforms, approximating multi-phase genesis, formation, preservation, and timescale.

Our observations of the form and content of the fluvial ridges suggest that the sandy gravel deposits stacked in the fluvial ridges were accumulated in paleochannels as ephemeral streams during high energy events over fast-responding landscapes. The upstream fluvial ridges suggest that intense precipitation events formed them through relatively high transportation rates from local sources, as inferred from the large median grain size ($D_{50} = 21-64$ mm), and the stacked gravels in the ridges lack internal stratification. Non-layered (depositionally structureless) gravels lacking bedding are common in rivers worldwide, and they have been explained by rapid transportation rates that reduce the chance of the development of an armored layer, particularly under ephemeral conditions (Laronne et al., 1994). Our estimates of paleodischarge using width relationships falling in the range of 33.4 ± 9.7 m³/s to 245 ± 71 m³/s also support the high energetic flows, particularly those sourced from relatively small drainage areas. These observations and calculations are consistent with the rainfall intensities involved in forming these rivers that fall in the range of 55-80 mm/hr over timescales ranging from 10^3 to 5 \times 10³ years (Zaki, King, et al., 2021). Moreover, when comparing the rivers with nearly similar discharge from different fluvial depositions environments, our drainage areas are close to those formed under flashier climates (Figure 8). A similar paleoenvironmental interpretation has been observed from fluvial ridge systems in the Kumtagh Desert, China (Wang et al., 2015), Ad Dwasir, Saudi Arabia (Matter et al., 2016), and the Atacama Desert (e.g., Morgan et al., 2014; Williams et al., 2021), where coarse materials from local bedrock sources are prevalent and were deposited over short durations, that is, thousands to tens of thousands years. Hayden, Lamb, and McElroy (2021) calculated the median of intermittency of the gravel-bed

river at 0.092, which means that the formative fluvial events could occur over a few days to several weeks every year, which could be applied to the ancient gravel-bed rivers investigated here.

The martian surface preserves multiple examples that record ancient rivers and streams are characterized by short-distance (a few kilometers from the source to the sink), single-thread channels like those in the western exposures of the Medusae Fossae Formation and on the plateau immediately west of Juventae Chasma on Mars (e.g., Di Pietro et al., 2018; Harrison et al., 2013; Weitz et al., 2008; Williams et al., 2007), reflecting river channel fill rather than channel-belt accumulation like the Izola outcrop in the northwestern shoulder of the Hellas Basin (Salese et al., 2020). Our interpretation based on the terrestrial analogs, therefore, lends support to the conclusion that the single-thread, short-distance source-to-sink fluvial ridges on Mars might have resulted from local bedrock sources subject to intense rainfall, leading to high sediment-transport rates.





Figure 12. Sediment accumulation rates within four paleorivers throughout the past 20 ka. The figure shows how sedimentation rates responded to the significant increase of rainfall in Fig (3).

Our preliminary investigation of the fan-shaped deposits shows that they record delta progradation and retrogradation at multiple stages with falling and rising the lake level (Figure 12). This progradation and retrogradation also could imply the presence of an ephemeral water body with fluctuating water levels (Saez et al., 2007). However, further age constraints and a detailed investigation of the internal architecture of these fans are essential to confirm the lake level fluctuations hypothesis. Levee forms, coupled with crevasse splays, have been observed at many fans, supporting the episodes of intense floods (e.g., Lepre, 2017), consistent with the observation of large clasts within the ridges. There are no constraints on the deposition of these fans, but Nicoll (1998) proposed the closed basin that hosts the fans to be one of these playa lakes that were active during the last AHP (early to mid-Holocene). To understand how sustained the fluvial activity lasted during the AHP, our estimates of migration rates and the time span required to form the meandering river systems within the deltas suggest at least 25.2 ± 0.9 up to 187 ± 6.2 years of river activity during the AHP. These durations indicate that these fluvial forms reached the bankfull channel for 1.2-8.8 days per year.

These ancient fans are also vital as a proxy for our understanding of early martian climate, since Mars has numerous fan-shaped landforms deposited in low-relief areas, bearing signatures of paleolakes based on existing outlet valleys and associated sedimentary rock (e.g., Fassett & Head, 2008;

Grotzinger et al., 2015). The accumulated sediments in the paleolakes as relief-inverted, fan-shaped deposits have been considered to be deltas by multiple studies (e.g., Di Achille & Hynek, 2010; Fassett & Head, 2005; Goudge et al., 2018; Malin & Edgett, 2003). Many of these deltas are locally sourced, recording deposition in shallow lakes (e.g., Di Achille & Hynek, 2010; Irwin et al., 2015; Malin & Edgett, 2003). Our preliminary geomorphic observations from the Saharan deltaic features, coupled with migration rate estimates and chronological context of the Saharan pluvial periods, suggest that similar fans on Mars might have formed under relatively wet conditions that spanned tens to a few hundred years over a total duration of thousands of years, in agreement with



Figure 13. Ridge thickness and calculated erosion rates for the six paleoriver sites; Gabal Hamam I and II, Gabal Masmas, Gabal El Sadd, Arqin I and II. Each point represents the mean ridge thickness and the mean erosion rate in each site. The mean erosion rate in the whole region is $\sim 1.1 \pm 0.2$ mm/yr.





Figure 14. (a) Digital elevation model derived from ALOS PALSAR data shows the distribution of the deltaic features and their catchment areas. (b) Deltaic deposit elevations suggest that the progradation of most deltas took place between 140 and 145 m. (c) Long profiles of the main trunk of the rivers that drained into Kiseiba-Tushka lake and built the fans, indicating that the accumulation of sediments occurred at multiple lake levels. The twenty rivers have been plotted in Figure S2 in Supporting Information S1.

current literature (e.g., Kleinhans, 2005; Lapôtre & Ielpi, 2020; Stucky de Quay et al., 2019). Such estimates suggest rapid sediment aggradation and burial, making the deltas an instructive site for testing if life ever existed on early Mars, as they may preserve organic matter because of rapid burial (Ehlmann et al., 2008; Goudge et al., 2018; Lapôtre & Ielpi, 2020). Furthermore, the fan deposits would have concentrated fine-grained materials that are favorable to preserving organic materials (Mangold et al., 2020).

Because of the possibility of organics being carried by water, fluvial deposits have been considered as a possible biosignature repository on Mars (Summons et al., 2011). This information is encoded in both the modes and durations of depositional history. Hayden et al. (2019) suggested that ridges recording long-lived fluvial activity that persisted over millions of years may favor habitability on early Mars. However, Williams et al. (2021) reported fluvial ridges from the Atacama Desert, Chile, preserving potential biosignatures, including plants and subsurface organisms over short durations, that is, thousands of years. This study also indicates that fluvial landforms deposited by short-duration fluvial activity (10–100s years over thousands of years) also supported





Figure 15. (a) and (b) Satellite image, coupled with sketch maps of main deltaic features showing the distribution of the major distributary channels, the point bars, and a crevasse splay. The satellite images were derived from Pleiades Satellite Imagery (© CNES, 2018; Distribution Airbus DS; ID: DS_PHR1A_201806180841178_FR1_PX_E030N23_1001_02095; acquisition date: 18/06/2018); 22°49'13.24"N; 30°47'11.82"E.

habitable environments for biosignature, including bacterial structures and organisms in the eastern Sahara (e.g., Cremaschi et al., 2010; Nicoll & Sallam, 2017). This suggestion makes such sediments scientifically very important for future sample return missions.

Because detailed measurements of sedimentary structures within the martian ridges are not feasible using orbital images, improving paleo-hydraulic methods that depend on ridge widths is important to constrain the paleodischarge better. Our estimates from the width-based discharge fall in the range of $33-245 \text{ m}^3$ /s. These estimates differ from discharge quantified from grain-size distribution and paleochannel channel geometry (43–138 m³/s; Zaki, King, et al., 2021). When comparing the two values quantified from the two methods, the width-discharge relationship yields a factor of ~70% higher than the discharge estimated from the grain size and slope reconstruction. The low discharge rate quantified from grain-size distribution and paleochannel geometry could be the minimum rate since there are uncertainties on grain-size measurements, and the erosion has played an important role in eroding the original channel geometry. However, this value could help when estimating the discharge based on width from orbital images.

After the termination of the last pluvial period in the Sahara at ~5.2 ka, the region dried up, and the aeolian processes, including deflation and wind abrasion (including occasional flashy rainfall events; Figure 18), lowered the surface at least 1.5–13.2 m, providing sediment some of which accumulated as various forms of dunes. Ridge thickness coupled with the age aridification enabled us to quantify the minimum erosion rates and denudation volumes involved in preserving the fluvial systems in the region. The calculated erosion rates are in the range of 0.3 ± 0.1 to 2 ± 0.3 mm/yr⁻¹, and the denudation volume of the eroded sediments is ~96 ± 24 × 10¹⁰ m³. These





Figure 16. (a) Well-preserved fan-shaped deposits in relief inversion setting within Kiseiba-Tushka paleolake. (b), (c), and (d) Close-up images show the main geomorphic structure within the deltaic features, including meanders, point bar, levees, and crevasse splay. North is up in all the images. The images were obtained from Pleiades Satellite Imagery (© CNES, 2018, Distribution Airbus DS; ID: DS_PHR1A_201806180841178_FR1_PX_E030N23_1001_02095).

volumes are nearly three times more than those eroded from the inverted fans in Oman (Mazeiles, 1987). The wide range of erosion rates (0.3–2 mm/yr) might be relevant to the consolidation control (Rohrmann et al., 2013). The consolidation variability of the Quaternary deposits that were never buried nor exhumed differs significantly from the Nubia Formation that buried, lithified, then exhumed. Similar ridges and modeled ages from crater counting can be used on the martian surface as a proxy to estimate the minimum erosion/exhumation rates, providing a better understanding of landscape evolution since the arid conditions of modern Mars replaced wet conditions that prevailed on early Mars.





Figure 17. Time series of two highly fluctuating modern lakes from the Egyptian Sahara showing deltas' formation over a few decades and confirming that tens of years are sufficient to form similar deltas to those investigated here (images source: Google Earth).

Overall, the AHP event offers a time-bounded, well-constrained narrative of environmental and climatic changes from wet to arid conditions that may exemplify the dramatic climate changes that Mars experienced in its early history. Our investigation of locally sourced fluvial depositional systems developed during the AHP provides glimpses into the formation and preservation of similar forms on Mars, suggesting that local and transient fluvial activity might have punctuated early martian climate over short-duration events. Alongside the martian ridges that might record fluvial activity over prolonged periods of geologic time ($\sim 10^5-10^6$ years; e.g., Davis et al., 2019; Hayden et al., 2019; Balme et al., 2020; Cardenas et al., 2020; Hayden & Lamb, 2020; Stucky de Quay et al., 2019), our results show that other ridges on Mars may have formed in brief, ephemeral events, over periods as short as a few thousand years. We emphasize that caution should be used when interpreting fluvial





Figure 18. Schematic diagram showing the transition from fluvial activity to aeolian processes in southern Egypt.

landforms from orbit due to this convergence of form. Taken together, our results add to the growing recognition of the potential of fluvial depositional systems for constraining the water and wind driven activities that might have dominated on early Mars.

6. Conclusions

The eastern Sahara shows a series of locally sourced, fluvial depositional systems, now expressed as sinuous to branching ridges in the modern landscape. They are widely distributed across \sim 38,000 km², extend for tens of kilometers, are as much as 65 m wide, and as high as 13 m. The ridges are discontinuous, representing fragments of dendritic drainage patterns. The internal structure of these ridges records two main stratigraphic units: Cretaceous Nubia Formation accumulated by fluvial activity, overlain by cemented sandy gravels fluvially deposited during mostly the AHP. The AHP gravel beds do not exceed 50% of the whole ridge thickness. Internally, the structure of the gravel beds displays non-layered gravels with angular to subrounded clasts ranging from 21 to 64 mm in grain size (D_{50}). The architecture of the gravel beds suggests that they were deposited by ephemeral rivers formed in response to intense precipitation and rapid sedimentation rates over rapidly responding land-scapes throughout short durations (10–100s years over ~10,000 years).

Some of the fluvial ridges are well preserved within fan-shaped deposits that we interpret as deltaic features. The minimum duration required to form the meandering system within the deltas spanned \sim 25–187 years, most likely were active throughout the AHP for \sim 10,000 years. The deltas were sourced from local small drainage areas, and their formation took place during at least three stages of lake-level fall and rise. Further investigation of the distribution, sedimentary architecture, age constraints of these fans would offer the possibility of understanding both the past climatic conditions that persisted during the Saharan pluvial periods.

The fluvial ridges have been eroded by both wind and water, forming ridges; we estimate the minimum erosion rates to be in the range of 2 ± 0.3 mm/y to 0.3 ± 0.1 mm/y. Our best estimate for the volume of eroded sediments is 95,000 \pm 25,000 km³; some of that volume has been reworked by the wind to form various forms of dunes, including linear, lee, and barchan dunes.

Locally sourced martian fluvial ridges that record short distance source-to-sink channels might have been formed from ephemeral streams fed by intense fluvial activity that took place over short durations, that is, thousands to a few tens of thousands of years. Our observations may imply that the early martian climate could have been



punctuated by short periods of locally high precipitation that formed short source-to-sink ephemeral channels and rivers. Partial erosion of such ridges during the transition from wet to dry conditions might have provided sufficient sediment to contribute to the formation of the aeolian dunes on Mars.

Data Availability Statement

Supporting information includes Figure S1 in Supporting Information S1 processed from Landsat 8 satellite imagery, accessible from the following website (Irons et al., 2012; https://earthexplorer.usgs.gov/). Table S1 includes raw data collected in the field of fluvial ridge locations, measurements, and discharge calculations. Tables S1, S2, and S3 in Supporting Information S1 represent measurements obtained from remotely sensed data, including Esri WorldImagery (https://www.arcgis.com/home/), and ALOS PALSAR DEM (Rosenqvist et al., 2007) archived here (Zaki, 2021; https://doi.org/10.5281/zenodo.6319858). The CTX and HiRISE images (Malin et al., 2007; McEwen et al., 2007) that have been used in preparing Figure 1 are also archived here (Zaki, 2021; https://doi.org/10.5281/zenodo.6319858). The Movies S1 and S2 in Supporting Information S1 were created via images from Google Earth historical imagery (https://earthengine.google.com/timelapse/). Pléiades' images (Lebegue et al., 2012) used in this study are not publicly available for commercial reasons but can be purchased from the following website (https://www.intelligence-airbusds.com/imagery/constellation/pleiades/).

References

- Abotalib, A. Z., Sultan, M., & Elkadiri, R. (2016). Groundwater processes in Saharan Africa: Implications for landscape evolution in arid environments. *Earth-Science Reviews*, 156, 108–136. https://doi.org/10.1016/j.earscirev.2016.03.004
- Balme, M. R., Gupta, S., Davis, J. M., Fawdon, P., Grindrod, P. M., Bridges, J. C., et al. (2020). Aram Dorsum: An extensive mid-Noachian age fluvial depositional system in Arabia Terra, Mars. *Journal of Geophysical Research: Planets*, 125(5), e2019JE006244. https://doi. org/10.1029/2019JE006244
- Berger, A., & Loutre, M. F. (1991). Insolation values for the climate of the last 10 millions years. *Quaternary Science Reviews*, 10(4), 297–317. https://doi.org/10.1016/0277-3791(91)90033-Q
- Blanchet, C. L., Osborne, A. H., Tjallingii, R., Ehrmann, W., Friedrich, T., Timmermann, A., et al. (2021). Drivers of river reactivation in North Africa during the last glacial cycle. *Nature Geoscience*, 14(2), 97–103. https://doi.org/10.1038/s41561-020-00671-3
- Burr, D. M., Williams, R. M. E., Wendell, K. D., Chojnacki, M., & Emery, J. P. (2010). Inverted fluvial features in the Aeolis/Zephyria Plana region, Mars: Formation mechanism and initial paleodischarge estimates. *Journal of Geophysical Research*, 115(E7), E07011. https://doi. org/10.1029/2009JE003496
- Cardenas, B. T., Mohrig, D., Goudge, T. A., Hughes, C. M., Levy, J. S., Swanson, T., et al. (2020). The anatomy of exhumed river-channel belts: Bedform to belt-scale river kinematics of the Ruby Ranch Member, Cretaceous Cedar Mountain Formation. *Sedimentology*, 67, 3655–3682. https://doi.org/10.1111/sed.12765
- Cardenas, N. T., Mohrig, A., & Goudge, T. A. (2017). Fluvial stratigraphy of valley fills at Aeolis Dorsa, Mars: Evidence for base-level fluctuations controlled by a downstream water body. GSA Bulletin, 130(3–4), 484–498. https://doi.org/10.1130/B31567.1
- Church, M., & Rood, K. (1983). Catalogue of alluvial river channel regime data (p. 99). The University of British Columbia, Department of Geography.
- Clarke, J., Pain, C. F., & Rupert, S. (2020). Complex expressions of inverted and exhumed relief in central Utah, and some martian counterparts. *Physical Geography*, 1–18. https://doi.org/10.1080/02723646.2020.1839161
- Cremaschi, M., Zerboni, A., Spotl, C., & Felletti, F. (2010). The calcareous tufa in the Tadrart Acacus Mt. (SW Fezzan, Libya): An early Holocene palaeoclimate archive in the central Sahara. *Palaeogeography, Palaeoclimatology, Palaeoecology, 287*(1–4), 81–94. https://doi.org/10.1016/j. palaeo.2010.01.019
- Davis, J. M., Balme, M., Grindrod, P. M., Williams, R. M. E., & Gupta, S. (2016). Extensive Noachian fluvial systems in Arabia Terra: Implications for early martian climate. *Geology*. https://doi.org/10.1130/G38247.1
- Davis, J. M., Gupta, S., Balme, M., Grindrod, P. M., Fawdon, P., Dickeson, Z. I., & Williams, R. M. E. (2019). A diverse array of fluvial depositional systems in Arabia Terra: Evidence for mid-Noachian to early Hesperian rivers on Mars. *Journal of Geophysical Research: Planets*, 124(7), 1913–1934. https://doi.org/10.1029/2019JE005976
- de Menocal, P., Ortiz, J., Guilderson, T., Adkins, J., Sarnthein, M., Baker, L., & Yarusinsky, M. (2000). Abrupt onset and termination of the african Humid Period: Rapid climate responses to gradual insolation forcing. *Quaternary Science Reviews*, 19(1–5), 347–361. https://doi. org/10.1016/S0277-3791(99)00081-5
- Di Achille, G., & Hynek, B. M. (2010). Ancient ocean on Mars supported by global distribution of deltas and valleys. *Nature Geoscience*, 3(7), 459–463. https://doi.org/10.1038/ngeo891
- Dickson, J. L., Lamb, M. P., Williams, R. M. E., Hayden, A. T., & Fischer, W. W. (2020). The global distribution of depositional rivers on early Mars. Geology, 49(5), 504–509. https://doi.org/10.1130/G48457.1
- Di Pietro, I., Ori, G. G., Pondrelli, M., & Salese, F. (2018). Geology of Aeolis Dorsa alluvial sedimentary basin, Mars. Journal of Maps, 14(2), 212–218. https://doi.org/10.1080/17445647.2018.1454350
- Eaton, B. C. (2013). 9.18 hydraulic geometry: Empirical investigations and theoretical approaches. In J. F. Shroder(Ed.), *Treatise on geomorphology*. Academic Press. (pp. 313–329).
- Ehlmann, B. L., Mustard, J. F., Fassett, C. I., Schon, S. C., Head, J. W., III., Des Marais, D. J., et al. (2008). Clay minerals in delta deposits and organic preservation potential on Mars. *Nature Geoscience*, 1(6), 355–358. https://doi.org/10.1038/ngeo207
- Embabi, N. S. (2018). Climatic conditions: Present and past: In Embabi (Ed.), *Landscapes and landforms of Egypt*. Springer International Publishing, (pp. 25–36). https://doi.org/10.1007/978-3-319-65661-8_3

Acknowledgments

We thank Barbara Tewksbury for her comments/suggestions that profoundly improved the paper. We thank Seán Doran for permitting us to use Figure 1d in this manuscript. We thank Hossam Khairy, Salah Ahmed, Saleh A. Eltayeb, and Mostafa El-Wakil for field assistance. Abdallah Zaki gratefully acknowledges the support of the Swiss Confederation excellence fellowships program (fellowship No: 2017.1006). The field research was sponsored by an Augustin Lombard fellowship and an Ernst et Lucie Schmidheiny Foundation fellowship. JMD gratefully acknowledges UK Space Agency funding (ST/R002355/1; ST/ V002678/1; ST/W002566/1). Pléiades was supported by public funds received in the framework of GEOSUD, a project (ANR-10-EQPX-20) of the program "Investissements d'Avenir" managed by the French National Research Agency. Alberto G. Fairén and F. Salese are supported by "MarsFirstWater," funded by the European Research Council, Consolidator Grant No. 818602. Finally, the manuscript has benefited from the constructive comments and reviews of Bradley J. Thomson and two anonymous reviewers. Open access funding provided by Universite de Geneve.

- Fassett, C. I., & Head, J. W. (2008). Valley network-fed, open-basin lakes on Mars: Distribution and implications for Noachian surface and subsurface hydrology. *Icarus*, 198(1), 37–56. https://doi.org/10.1016/j.icarus.2008.06.016
- Fassett, C. I., & Head, J. W., III. (2005). Fluvial sedimentary deposits on Mars: Ancient deltas in a crater lake in the Nili Fossae region. Geophysical Research Letters, 32(14). https://doi.org/10.1029/2005GL023456
- Foix, N., Allard, J. O., Paredes, J. M., & Giacosa, R. E. (2012). Fluvial styles, palaeohydrology and modern analogues of an exhumed, cretaceous fluvial system: Cerro Barcino formation, cañadón asfalto basin, Argentina. Cretaceous Research, 34, 298–307. https://doi.org/10.1016/j. cretres.2011.11.010
- Giegengack, R. F., & Zaki, A. S. (2017). Inverted topographic features, now submerged beneath the water of Lake Nasser, document a morphostratigraphic sequence of high-amplitude late-Pleistocene climate oscillation in Egyptian Nubia. *Journal of African Earth Sciences*, 136, 176–187. https://doi.org/10.1016/j.jafrearsci.2017.06.027
- Goudge, T. A., Head, W. H., Mustard, J. F., & Fassett, C. I. (2012). An analysis of open-basin lake deposits on Mars: Evidence for the nature of associated lacustrine deposits and post-lacustrine modification processes. *Icarus*, 2179(1), 211–229. https://doi.org/10.1016/j.icarus.2012.02.027
 Goudge, T. A., Mohrig, D., Cardenas, B. T., Hughes, C. M., & Fasset, C. I. (2018). Stratigraphy and paleohydrology of delta channel deposits.
- Jezero crater, Mars. *Icarus*, 301, 58–75. https://doi.org/10.1016/j.icarus.2017.09.034
- Grotzinger, J. P., Gupta, S., Malin, M. C., Rubin, D. M., Schieber, J., Siebach, K., et al. (2015). Deposition, exhumation, and paleoclimate of an ancient lake deposit, Gale Crater, Mars. Science, 350(6257). https://doi.org/10.1126/science.aac7575
- Gulick, V. C. (2001). Origin of the valley networks on Mars: A hydrological perspective. *Geomorphology*, 37(3–4), 241–268. https://doi.org/10.1016/S0169-555X(00)00086-6
- Hack, J. T. (1957). Studies of Longitudinal Stream Profiles in Virginia and Maryland (Vol. 294, pp. 1–97). US Geological Survey Professional Paper, B.
- Hamdan, M. A., Refaat, A. A., & Abdel Wahed, M. (2016). Morphologic charcteristics and migration rate assessment of barchan dunes in the southeastern Western Desert of Egypt. *Geomorphology*, 257, 57–74. https://doi.org/10.1016/j.geomorph.2015.12.026
- Hamdan, M. A., Refaat, A. A., Abu Anwar, E., & Shallaly, N. A. (2015). Source of the aeolian dune sand of Toshka area, southeastern Western Desert, Egypt. Aeolian Research, 17, 275–289. https://doi.org/10.1016/j.aeolia.2014.08.005
- Harrison, S. K., Balme, M. R., Hagermann, A., Murray, J. B., Muller, J. P., & Wilson, A. (2013). A branching, positive relief network in the middle member of the Medusae Fossae Formation, equatorial Mars—evidence for sapping? *Planetary and Space Science*, 85, 142–163. https:// doi.org/10.1016/j.pss.2013.06.004
- Hayden, A. T., & Lamb, M. P. (2020). Fluvial sinuous ridges of the Morrison Formation, USA: Meandering, scarp retreat, and implications for Mars. Journal of Geophysical Research: Planets, 125(10), e2020JE006470. https://doi.org/10.1029/2020JE006470
- Hayden, A. T., Lamb, M. P., Fischer, W. W., Ewing, R. C., McElory, B. J., & Williams, R. M. E. (2019). Formation of sinuous ridges by inversion of river-channel belts in Utah, USA, with implications for Mars. *Icarus*, 332, 92–110. https://doi.org/10.1016/j.icarus.2019.04.019
- Hayden, A. T., Lamb, P. L., & Carney, A. J. (2021). Similar curvature-to-width ratios for channels and channel belts: Implications for paleo-hydraulics of fluvial ridges on Mars. *Geology*, 49(7), 837–841. https://doi.org/10.1130/G48370.1
- Hayden, A. T., Lamb, P. L., & McElroy, B. J. (2021). Constraining the timespan of fluvial activity from the intermittency of sediment transport on Earth and Mars. *Geophysical Research Letters*, 48(16). https://doi.org/10.1029/2021GL092598
- Haynes, C. V., Eyles, C. H., Pavlish, L. A., Ritchie, J. C., & Rybak, M. (1989). Holocene palaeoecology of the eastern Sahara; Selima Oasis. *Quaternary Science Reviews*, 8(2), 109–136. https://doi.org/10.1016/0277-3791(89)90001-2
- Hoelzmann, P., Kruse, H., & Rottinger, F. (2000). Precipitation estimates for the eastern Saharan palaeomonsoon based on a water balance model of the West Nubian Palaeolake Basin. *Global and Planetary Change*, 26(1–2), 105–120. https://doi.org/10.1016/S0921-8181(00)00038-2
- Ielpi, A., & Ghinassi, M. (2014). Planform architecture, stratigraphic signature and morphodynamics of an exhumed Jurassic meander plain (Scalby Formation, Yorkshire, UK). Sedimentology, 61(7), 1923–1960. https://doi.org/10.1111/sed.12122
- Irons, J. R., Dwyer, J. L., & Barsi, J. A. (2012). The next Landsat satellite: The Landsat data continuity mission. Remote Sensing of Environment, 122, 11–21. https://doi.org/10.1016/j.rse.2011.08.026
- Irwin, R. P., Lewis, K. W., Howard, A. D., & Grant, J. A. (2015). Paleohydrology of Eberswalde crater, Mars. Geomorphology, 240, 83–101. https://doi.org/10.1016/j.geomorph.2014.10.012
- Jacobsen, R. E., & Burr, D. M. (2018). Errors in Martian paleodischarges skew interpretations of hydrologic history: Case study of the Aeolis Dorsa, Mars, with insights from the Quinn River, NV. Icarus, 302, 407–714. https://doi.org/10.1016/j.icarus.2017.11.014
- Kite, E. S. (2018). Geologic constraints on early Mars climate. Space Science Reviews, 210(1), 10. https://doi.org/10.1007/s11214-018-0575-5
- Kite, E. S., Howard, A. D., Lucas, A. S., & Lweis, K. W. (2015). Resolving the era of river-forming climates on Mars using stratigraphic logs of river-deposit dimensions. *Earth and Planetary Science Letters*, 420, 55–65. https://doi.org/10.1016/j.epsl.2015.03.019
- Kite, E. S., Mayer, D. P., Wilson, S. A., Davis, J. M., Lucas, A. S., & Stucky de Quay, G. (2019). Persistence of intense, climate-driven runoff late in Mars history. *Science Advances*, 5(3), eaav7710. https://doi.org/10.1126/sciadv.aav7710
- Kleinhans, M. G. (2005). Flow discharge and sediment transport models for estimating a minimum timescale of hydrological activity and channel and delta formation on Mars. *Journal of Geophysical Research*, 110(E12), E12003. https://doi.org/10.1029/2005JE002521
- Kuper, R., & Kropelin, S. (2006). Climate-controlled Holocene occupation in the Sahara: Motor of Africa's evolution. *Science*, 303(5788), 803–807. https://doi.org/10.1126/science.1130989
- Lammer, H., Chassefière, E., Karatekin, Ö., Morschhauser, A., Niles, P. B., Mousis, O., et al. (2013). Outgassing history and escape of the Martian atmosphere and water inventory. *Space Science Reviews*, 174(1–4), 113–154. https://doi.org/10.1007/s11214-012-9943-8
- Lapôtre, M., & Ielpi, A. (2020). The pace of fluvial meanders on Mars and implications for the Western Delta deposits of Jezero Crater. AGU Advances, 1(2), e2019AV000141. https://doi.org/10.1029/2019AV000141
- Laronne, J. B., Reid, I., Yitshak, Y., & Frostick, L. E. (1994). The non-layering of gravel streambeds under ephemeral flood regimes. *Journal of Hydrology*, 159(1–4), 353–363. https://doi.org/10.1016/0022-1694(94)90266-6
- Lebegue, L., Greslou, D., de Lussy, F., Fourest, S., Blanchet, G., Lacherade, L., et al. (2012). PLEIADES-HR image quality commissioning. International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences, 1, 561–566. https://doi.org/10.5194/ isprsarchives-XXXIX-B1-561-2012
- Lepre, C. J. (2017). Crevasse-splay and associated depositional environments of the hominin-bearing lower okote member, Koobi Fora Formation (plio-pleistocene), Kenya. *The Depositional Record*, 3(2), 161–186. https://doi.org/10.1002/dep2.31
- Mackenzie, W. S., Adams, A. E., & Brodie, K. H. (2017). Rocks and minerals in thin section (p. 232). CRC Press.
- Madof, A. S., Bertoni, C., & Lofi, J. (2019). Discovery of vast fluvial deposits provides evidence for drawdown during the late Miocene Messinian Salinity Crisis. *Geology*, 47(2), 171–174. https://doi.org/10.1130/G45873.1
- Maizels, J. (1983). Paleovelocity and paleodischarge determination for coarse gravel deposits. In K. J. Gregory(Ed.), *Background to paleohydrology a perspective* (pp. 101–139). John Wiley.

- Maizels, J. (1987). Plio-Pleistocene raised channel systems of the Western Sharqiya (Wahiba), Oman. In L. Frostick, & I. Reid(Eds.), Desert sediments-ancient and modern (Vol. 35, pp. 31–50). British Geological Society, Special Publication. https://doi.org/10.1144/GSL. SP.1987.035.01.04
- Maizels, J. (1990). Raised channels as indicators of palaeohydrologic change-a case study from Oman. *Palaeogeography, Palaeoclimatology, Palaeoecology*, 76(3–4), 241–277. https://doi.org/10.1016/0031-0182(90)90115-N
- Malin, M. C., Bell, J. F., III., Cantor, B. A., Caplinger, M. A., Calvin, W. M., Clancy, R. T., et al. (2007). Context Camera investigation on board the Mars reconnaissance orbiter. *Journal of Geophysical Research*, 112(E5), E05S04. https://doi.org/10.1029/2006JE002808
- Malin, M. C., & Edgett, K. S. (2003). Evidence for persistent flow and aqueous sedimentation on early Mars. Science, 302(5652), 1931–1934. https://doi.org/10.1126/science.1090544
 - Mangold, N., Dromart, G., Ansan, V., Salese, F., Kleinhans, M. G., Massé, M., et al. (2020). Fluvial regimes, morphometry, and age of Jezero crater paleolake inlet valleys and their exobiological significance for the 2020 Rover Mission Landing Site. Astrobiology, 20(8), 994–1013. https://doi.org/10.1089/ast.2019.2132
- Mangold, N., Gupta, S., Gasnault, O., Dromart, G., Tarnas, J. D., Sholes, S. F., et al. (2021). Perseverance rover reveals an ancient delta-lake system and flood deposits at Jezero crater, Mars. Science, 374(6568), 711–717. https://doi.org/10.1126/science.abl4051
- Mangold, N., Quantin, C., Ansan, V., Delacourt, C., & Allemand, P. (2004). Evidence for precipitation on Mars from dendritic valleys in the Valles Marineris area. Science, 305(5680), 78–81. https://doi.org/10.1126/science.1097549
- Martrat, B., Frimalt, J. O., Lopez-Martines, C., Cacho, I., Sierro, F. J., Flores, J. A., et al. (2004). Abrupt temperature changes in the Western Mediterranean over the past 250,000 years. *Science*, 306(5702), 1762–1765. https://doi.org/10.1126/science.1101706
- Matter, A., Mahjoub, A., Neubert, E., Preusser, F., Schwalb, A., Szidat, S., & Wulf, G. (2016). Reactivation of the Pleistocene trans-Arabian Wadi ad Dawasir fluvial system (Saudi Arabia) during the Holocene humid phase. *Geomorphology*, 270, 88–101. https://doi.org/10.1016/j. geomorph.2016.07.013
- Maxwell, T. A., Issawi, B., & Haynes, C. V. (2010). Evidence for Pleistocene lakes in the Tushka region, south Egypt. Geology, 38(12), 1135– 1138. https://doi.org/10.1130/G31320.1
- McEwen, A. S., Eliason, E. M., Bergstrom, J. W., Bridges, N. T., Hansen, C. J., Delamere, W. A., et al. (2007). Mars reconnaissance orbiter's high resolution imaging science experiment (HiRISE). *Journal of Geophysical Research*, 112, E05S02. https://doi.org/10.1029/2005JE002605 McKee, E. D. (1963). Origin of the Nubian and similar sandstone. *Geologische Rundschau*, 52(2), 551–587. https://doi.org/10.1007/bf01821143
- Morgan, A. M., Howard, A. D., Hobley, D. E. J., Moore, J. M., Dietrich, W. E., Williams, R. M. E., et al. (2014). Sedimentology and climatic environment of alluvial fans in the Martian Saheki crater and a comparison with terrestrial fans in the Atacama Desert. *Icarus*, 229, 131–156. https://doi.org/10.1016/j.icarus.2013.11.007
- Nicoll, K. (2004). Recent environmental change and prehistoric human activity in Egypt and Northern Sudan. Quaternary Science Reviews, 23(5–6), 561–580, https://doi.org/10.1016/j.guascirey.2003.10.004
- Nicoll, K. A. (1998). Holocene playas as sedimentary evidence for recent climate change in the presently hyperarid western desert, Egypt. PhD thesis (p. 290). The University of Arizona. Retrieved from http://hdl.handle.net/10150/282835
- Nicoll, K. A., & Sallam, E. S. (2017). Paleospring tufa deposition in the Kurkur Oasis region and implications for tributary integration with the River Nile in southern Egypt. Journal of African Earth Sciences, 136, 239–251. https://doi.org/10.1016/j.jafrearsci.2016.10.014
- Oliver, J. (1965). Evaporation rates and rainfall regime in central and northern Sudan. Weather, 20(2), 58–64. https://doi.org/10.1002/j.1477-8696.1965.tb02147.x
- Pain, C. F., & Ollier, C. D. (1995). Inversion of relief A component of landscape evolution. Geomorphology, 12(2), 151–165. https://doi. org/10.1016/0169-555X(94)00084-5
- Palchan, D., & Torfstien, A. (2019). A Drop in Sahara Dust Fluxes Records the Northern Limits of the African Humid Period (Vol. 3803). Nature Communication. https://doi.org/10.1038/s41467-019-11701-z
- Patton, P. C., & Baker, V. R. (1976). Morphometry and floods in small drainage basins subject to diverse hydrogeomorphic controls. Water Resources Research, 12(5), 941–952. https://doi.org/10.1029/WR012i005p00941
- Ritchie, J. C., & Haynes, C. V. (1987). Holocene vegetation zonation in the eastern Sahara. Nature, 330(6149), 645–647. https://doi.org/10.1038/330645a0
- Rohrmann, A., Heemance, R., Kapp, P., & Cai, F. (2013). Wind as the primary driver of erosion in the Qaidam Basin, China. Earth and Planetary Science Letters, 374, 1–10. https://doi.org/10.1016/j.epsl.2013.03.011
- Rosenqvist, A., Shimada, M., Ito, N., & Watanabe, M. (2007). ALOS PALSAR: A Pathfinder mission for global-scale monitoring of the environment. *IEEE Transactions on Geoscience and Remote Sensing*, 45(11), 3307–3316. https://doi.org/10.1109/TGRS.2007.901027
- Sáez, A., Valero-Garcés, B. L., Moreno, A., Bao, R., Pueyo, J. J., González-Sampériz, P., et al. (2007). Lacustrine sedimentation in active volcanic settings: The late quaternary depositional evolution of Lake Chungará (Northern Chile). Sedimentology, 54(5), 1191–1222. https:// doi.org/10.1111/j.1365-3091.2007.00878.x
- Said, R., & Issawi, B. (1964). Preliminary results of a geological expedition to lower Nubia and to kurkur and dungul oases, Egypt: In Wendorf (Ed.), Contributions to the prehistory of Nubia, Fort Burgwin Research Center and Southern Methodist University Press, (pp. 1–28).
- Salese, F., McMahon, W. J., Balme, M. R., Ansan, V., Davis, J. M., & Kleinhans, M. G. (2020). Sustained fluvial deposition recorded in Mars' Noachian stratigraphic record. *Nature Communications*, 11(1), 2067. https://doi.org/10.1038/s41467-020-15622-0
- Salese, F., Pondrelli, M., Neeseman, A., Schmidt, G., & Ori, G. G. (2019). Geological evidence of planet-wide groundwater system on Mars. Journal of Geophysical Research: Planets, 124(2), 374–395. https://doi.org/10.1029/2018JE005802
- Schuster, M., Roquin, C., Duringer, P., Brunet, M., Caugy, M., Fontugne, M., et al. (2005). Holocene lake mega-Chad palaeoshorelines from space. *Quaternary Science Reviews*, 24(16–17), 1821–1827. https://doi.org/10.1016/j.quascirev.2005.02.001
- Som, S. M., Montgomery, D. R., & Greenberg, H. M. (2009). Scaling relations for large Martian valleys. Journal of Geophysical Research Planets, 114, E02005. https://doi.org/10.1029/2008JE003132
- Stucky de Quay, G., Kite, E. S., & Mayer, D. P. (2019). Prolonged fluvial activity from channel-fan systems on Mars. Journal of Geophysical Research: Planets, 124, https://doi.org/10.1029/2019JE006167
- Summons, R. E., Amend, J. P., Bish, D., Buick, R., Cody, G. D., Marais, D., et al. (2011). Preservation of martian organic and environmental records: Final report of the Mars Biosignature Working Group. Astrobiology, 11(2), 157–181. https://doi.org/10.1089/ast.2010.0506
- Tewksbury, B. J., Mehrtens, C. J., Steven, A., Gohlke, S. A., Tarabees, E. A., & Hogan, J. P. (2017). Constraints from Mesozoic siliciclastic cover rocks and satellite image analysis on the slip history of regional E-W faults in the southeast Western Desert, Egypt. Journal of African Earth Sciences, 136, 119–135. https://doi.org/10.1016/j.jafrearsci.2017.06.026
- Trampush, S. M., Huzurbazar, S., & McElroy, B. (2014). Empirical assessment of theory for bankfull characteristics of alluvial channels. Water Resources Research, 50(12), 9211–9220. https://doi.org/10.1002/2014wr015597

- Wagner, B., Vogel, H., Francke, A., Friedrich, T., Donders, T., Lacey, J. H., et al. (2019). Mediterranean winter rainfall in phase with African monsoons during the past 1.36 million years. *Nature*, 573(7773), 256–260. https://doi.org/10.1038/s41586-019-1529-0
- Wang, Z.-T., Lai, Z.-P., & Qu, J.-J. (2015). Inverted relief landforms in the Kumtagh Desert of northwestern China: A mechanism to estimate wind erosion rates. *Geological Journal*, 52(1), 131–140. https://doi.org/10.1002/gj.2739
- Weitz, C. M., Milliken, R. E., Grant, J. A., McEwen, A. S., Williams, R. M. E., & Bishop, J. L. (2008). Light-toned strata and inverted channels adjacent to Juventae and Ganges chasmata, Mars. *Geophysical Research Letters*, 35(19), L19202. https://doi.org/10.1029/2008GL035317
- Wendorf, F. A., & Schild, R. A. (1980). Prehistory of the eastern Sahara (p. 414p). Southern Methodist University.
- Williams, R. M. E. (2007). Global spatial distribution of raised curvilinear features on Mars. Lunar Planet. Science Conference 38th. Abstract # 1821.
- Williams, R. M. E., Chidsey, T. C., Jr., & Eby, D. E. (2007). Exhumed paleochannels in central Utah analogs for raised curvilinear features on Mars. In G. C. Willis, M. D. Hylland, D. L. Clark, & T. C. Chidsey Jr (Eds.), Central Utah – Diverse geology of a dynamic landscape, Utah Geological Association Publication. (Vol. 36, pp. 220–235).
- Williams, R. M. E., Irwin, R. P., Dobrea, E. Z. N., Howard, A. D., Dietrich, W. E., & Cawley, J. C. (2021). Inverted channel variations identified on a distal portion of a bajada in the Central ATACAMA desert, Chile. *Geomorphology*, 393, 107925. https://doi.org/10.1016/j. geomorph.2021.107925
- Williams, R. M. E., Irwin, R. P., & Zimbelman, J. R. (2009). Evaluation of paleohydrologic models for terrestrial inverted channels: Implications for application to Martian sinuous ridges. *Geomorphology*, 107(3–4), 300–315. https://doi.org/10.1016/j.geomorph.2008.12.015
- Wolman, M. G. A. (1954). Method of sampling coarse river-bed material. EOS, Transactions American Geophysical Union, 35(6), 951–956. https://doi.org/10.1029/TR035i006p00951
- Wordsworth, R. D. (2016). The climate of early Mars. Annual Review of Earth and Planetary Sciences, 44(1), 381–408. https://doi.org/10.1146/annurev-earth-060115-012355
- Zaki, A. S. (2021). Fluvial depositional systems of the african Humid Period: An analog for an early, wet Mars in the eastern Sahara. Zenodo. https://doi.org/10.5281/zenodo.63198
- Zaki, A. S., & Giegengack, R. (2016). Inverted topography in the southeastern part of the Western Desert of Egypt. Journal of African Earth Sciences, 121, 56–61. https://doi.org/10.1016/j.jafrearsci.2016.05.020
- Zaki, A. S., King, G. E., Haghipour, N., Giegengack, R., Watkins, S. E., Gupta, S., et al. (2021). Did increased flooding during the African Humid Period force migration of modern humans from the Nile Valley? *Quaternary Science Reviews*, 272. 107200. https://doi.org/10.1016/j. quascirev.2021.107200
- Zaki, A. S., Pain, C., Edgett, K. E., & Giegengack, R. (2018). Inverted stream channels in the Western Desert of Egypt: Synergistic remote, field observations and laboratory analysis on Earth with application to Mars. *Icarus*, 309, 105–124. https://doi.org/10.1016/j.icarus.2018.03.001
- Zaki, A. S., Pain, C., Edgett, K. S., & Castelltort, S. (2021). Global inventory of fluvial ridges on Earth, with lessons applicable to Mars. *Earth-Science Reviews*, 216 103561. https://doi.org/10.1016/j.earscirev.2021.103561